Author's response to reviews

Title: Patterns of Prescription Drug Expenditures and Medication Adherence Among Medicare Part D Beneficiaries with and without the Low-Income Supplement

Authors:

Stella M Yala MD (sylala@mednet.ucla.edu)
Norman Turk MS (NTurk@mednet.ucla.edu)
Susan Ettner PhD (settnr@mednet.ucla.edu)
Obidiugwu Kenrik Duru MD, MSHS (KDuru@mednet.ucla.edu)
Carol M Mangione MD, MSPH (CMangione@mednet.ucla.edu)
Arleen F Brown MD, PhD (abrown@mednet.ucla.edu)

Version: 3 Date: 23 July 2014

Author's response to reviews: see over
Dear Editors,

Thank you for this opportunity to resubmit our work. In response to the requested changes, we made the following changes to the manuscript:

<table>
<thead>
<tr>
<th>1) In the abstract the purpose of the study as well as the context needs to be included</th>
<th>The purpose and context of the study was added to the abstract section.</th>
</tr>
</thead>
<tbody>
<tr>
<td>2) At the beginning of the main text, after the abstract, the heading Background is needed</td>
<td>The heading Background was added.</td>
</tr>
<tr>
<td>3) A conclusion section is required following the discussion</td>
<td>A Conclusion section was added.</td>
</tr>
<tr>
<td>4) Please include line numbers</td>
<td>Page numbers were included</td>
</tr>
<tr>
<td>5) Authors’ Contributions and Conflict of Interest statements need formatting so that it conforms to our house style</td>
<td>These two sections were reformatted to comply.</td>
</tr>
</tbody>
</table>

Thank you,

O. Kenrik Duru for the authors
Associate Professor of Medicine, David Geffen School of Medicine, Los Angeles, CA
10940 Wilshire Blvd., Suite 700
Los Angeles, CA 90024
Phone (310) 794-8138
Fax (310) 794-0723
Email kduru@mednet.ucla.edu